Author's response to reviews

Title: Association between difficulty initiating sleep in older adults and the combination of leisure-time physical activity and consumption of milk and milk products: a cross-sectional study

Authors:

Naruki Kitano (n.kitano0911@gmail.com)
Kenji Tsunoda (k.tsunoda.taiiku@gmail.com)
Taishi Tsuji (tsuji.t.taishi@jyu.fi)
Yosuke Osuka (osuka@stat.taiiku.tsukuba.ac.jp)
Takashi Jindo (takashi.jindo@gmail.com)
Kiyoji Tanaka (tanaka@taiiku.tsukuba.ac.jp)
Tomohiro Okura (okura@taiiku.tsukuba.ac.jp)

Version: 2
Date: 12 May 2014

Author's response to reviews:

To whom it may concern,

I revised my manuscript and added the line numbering.
If you please, confirm the renewal manuscript.

We all authors believe that findings from this study will be of special interest to the readers of BMC Geriatrics.

Many thanks in advance for considering our manuscript.

Sincerely,

Naruki Kitano